Author's response to reviews

Title: Economics of implementation, a case study of QUERI: QUERI Series

Authors:
Mark W Smith (mark.smith9@va.gov)
Paul G Barnett (paul.barnett@va.gov)

Version: 6 Date: 18 April 2008

Author's response to reviews:

There were four changes:
* Title reworded to avoid redundancy after required phrase "QUERI Series" was added at the end
* Authors' qualifications removed
* Erroneous reference number corrected on p. 9: first reference to [26] changed to [25]* URL on page 15 deleted from text and added as new reference 47; all citations to later references updated with new numbers (old nos. 47-51 are now 48-52).